**Author’s response to reviews**

**Title:** Another Whipple’s Triad? Pericardial, Myocardial and Valvular Disease in an Unusual Case Presentation from a Canadian Perspective

**Authors:**

Christina Thornton (christina.thornton@ahs.ca)

Yinong Wang (Yinong.Wang@cls.ab.ca)

Martin Köbel (Martin.Koebel@cls.ab.ca)

Kathryn Bernard (kathy.bernard@canada.ca)

Tamara Burdz (tamara.burdz@canada.ca)

Andrew Maitland (Andrew.Maitland@albertahealthservices.ca)

Jose Ferraz (ferraz@ucalgary.ca)

Paul Beck (plbeck@ucalgary.ca)

Andre Ferland (Andre.Ferland@albertahealthservices.ca)

**Version:** 2 **Date:** 06 Nov 2019

**Author’s response to reviews:**

1. Please ensure the corresponding author, and their email address, is the same on both the submission system and on the title page of your manuscript.

   - This has been revised to be the same.

2. Please combine your Discussion and Conclusion into one section called 'Discussion and Conclusions' as per our submission guidelines.

   - This has been done on line 157.

3. Please include a ‘Consent for publication’ section under your Declarations, and copy your consent for publication statement under it.
4. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

-This has been done on line 228.

5. Please upload your multi-panel figures as one figure. I.e. each separate figure (figure 1, figure 2, etc) should be one file, not multiple files.

-This has been revised and uploaded.

6. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

-This has been done.